

REVIEW HISTORY

Paper: **“Flipped Learning for Repurposing EFL Class Time in Moroccan University”**

Corresponding Author: Mohamed Jaafari  
Email: [jaafarimd@yahoo.com](mailto:jaafarimd@yahoo.com)

Doi: 10.19044/ejes.v6no4a1

Peer review:  
Reviewer 1: Blinded

Reviewer 2: Blinded

Published: 29.12.2019